

This is a repository copy of *Pertuzumab* for the *Neoadjuvant Treatment* of *Early-Stage HER2-Positive Breast Cancer: An Evidence Review Group Perspective of a NICE Single Technology Appraisal*.

White Rose Research Online URL for this paper: <a href="https://eprints.whiterose.ac.uk/119850/">https://eprints.whiterose.ac.uk/119850/</a>

Version: Accepted Version

## Article:

Squires, H. orcid.org/0000-0002-2776-4014, Pandor, A., Thokala, P et al. (5 more authors) (2018) Pertuzumab for the Neoadjuvant Treatment of Early-Stage HER2-Positive Breast Cancer: An Evidence Review Group Perspective of a NICE Single Technology Appraisal. Pharmacoeconomics, 36 (1). pp. 29-38. ISSN 1170-7690

https://doi.org/10.1007/s40273-017-0556-7

# Reuse

Items deposited in White Rose Research Online are protected by copyright, with all rights reserved unless indicated otherwise. They may be downloaded and/or printed for private study, or other acts as permitted by national copyright laws. The publisher or other rights holders may allow further reproduction and re-use of the full text version. This is indicated by the licence information on the White Rose Research Online record for the item.

## **Takedown**

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



Pertuzumab for the neoadjuvant treatment of early stage HER2-positive breast cancer: An Evidence Review Group Perspective of a NICE Single Technology Appraisal

**Running title:** Neoadjuvant pertuzumab treatment for HER2-positive breast cancer: Evidence Review Group Perspective

Squires H<sup>1</sup>, Pandor A<sup>1</sup>, Thokala P<sup>1</sup>, Stevens JW<sup>1</sup>, Kaltenthaler E<sup>1</sup>, Clowes M<sup>1</sup>, Coleman R<sup>2</sup>, Wyld L<sup>2</sup>

Corresponding author: Hazel Squires, h.squires@sheffield.ac.uk, 0114 2220765

<sup>&</sup>lt;sup>1</sup> The School of Health and Related Research (ScHARR), University of Sheffield

<sup>&</sup>lt;sup>2</sup> Department of Oncology Cancer Clinical Trials Centre, Academic Unit of Clinical Oncology, Weston Park Hospital, Sheffield

<sup>&</sup>lt;sup>3</sup> Consultant and Senior Lecturer, Department of Oncology, The Medical School, Sheffield

#### Abstract

As part of its Single Technology Appraisal process, the National Institute for Health and Care Excellence (NICE) invited the manufacturer of pertuzumab (Perjeta®; Roche Products Limited) to submit evidence of its clinical and cost-effectiveness for the neoadjuvant treatment of women with high risk early stage, HER2positive breast cancer when used in combination with trastuzumab and chemotherapy. High risk women included those with locally advanced (including inflammatory) breast cancer and women with high risk early breast cancer (classified as T2/3 or N1). The School of Health and Related Research Technology Appraisal Group at the University of Sheffield was commissioned to act as the independent Evidence Review Group (ERG). This article presents the critical review of the company's submission by the ERG and the outcome of the NICE guidance. The clinical data were mainly taken from a Phase II, randomised, open-label, active controlled study (NeoSphere), which reported a significant advantage in terms of pathological complete response (pCR) rates of pertuzumab in combination with trastuzumab and chemotherapy, compared with trastuzumab alone with chemotherapy (45.8% versus 29.0%, p=0.0141). The company did not make any indirect comparisons. A metaanalysis of 12 neoadjuvant studies investigating the relationship between pCR and event-free survival (EFS) was used to extrapolate the outcomes reported in the NeoSphere study. A cardiac safety study (TRYPHAENA) demonstrated the safety of pertuzumab. The company undertook a model-based economic evaluation of neoadjuvant pertuzumab plus trastuzumab and docetaxel compared with neoadjuvant trastuzumab and docetaxel over a lifetime horizon from the National Health Service (NHS) and Personal Social Services (PSS) perspective. The probabilistic incremental cost-effectiveness ratio (ICER) was estimated to be £20,104 per quality-adjusted life-year (QALY) gained for pertuzumab alongside trastuzumab and docetaxel compared with trastuzumab and docetaxel, which was revised to £21,869 per QALY gained following the clarification process. The ERG corrected an error in the digitisation of the survivor functions and modified the clinically inappropriate assumption that recurrence is zero after 7 years. The ERG's probabilistic base case was £23,962 per QALY gained. During the appraisal, to mitigate the uncertainties associated with the evidence, the company offered a patient access scheme (PAS), which led to the NICE Appraisal Committee recommending pertuzumab in this patient group, subject to the company providing the agreed discount in the PAS.

# **Key points for decision makers**

- Neoadjuvant pertuzumab in combination with trastuzumab and chemotherapy provides a statistically significant advantage in terms of pathological complete response rates compared with trastuzumab and chemotherapy for treating HER2-positive breast cancer which is either locally advanced, inflammatory, or early stage (at a high-risk of recurrence).
- The safety of pertuzumab has been demonstrated.
- Given the patient access scheme proposed by the company, neoadjuvant pertuzumab is considered to be a cost-effective use of NHS resources.

# 1. Introduction

The National Institute for Health and Care Excellence (NICE) is an independent organisation whose responsibilities include providing national guidance to the National Health Service (NHS) in England on health technologies. The NICE Single Technology Appraisal (STA) process usually covers new health technologies

within a single indication, soon after UK market authorisation national [1]. The company submits evidence on the clinical and cost-effectiveness of the technology, including a *de novo* economic model, and an independent Evidence Review Group (ERG) reviews the submission. The NICE Appraisal Committee (AC) considers the evidence submitted by the company and the ERG, alongside testimony from experts and other stakeholders, in order to develop national recommendations for England. These findings are reported within a Final Appraisal Determination (FAD). An Appraisal Consultation Document (ACD) is initially produced if the recommendations from the AC are restrictive or additional clarification is required from the company about their submission. All stakeholders have an opportunity to comment on the ACD before the AC meets again to produce the FAD.

The School of Health and Related Research Technology Assessment Group (ScHARR-TAG) at the University of Sheffield were the independent Evidence Review Group (ERG) who produced a critical review of the company's submission. This article presents a summary of the ERG report at the time of the assessment and the outcome of the NICE guidance for the STA of neoadjuvant pertuzumab for treating patients with HER2-positive breast cancer who are at high risk of recurrence, when used in combination with trastuzumab and chemotherapy. This is one of a series of STA summaries being published in PharmacoEconomics. Full details of all relevant appraisal documents can be found on the NICE website [2].

# 2. The Decision Problem

Pertuzumab is licensed for use in patients with early stage HER-2 positive breast cancer at high risk of recurrence (locally advanced, inflammatory or early stage breast cancer), before the patient undergoes surgery, to be used in combination with trastuzumab and chemotherapy [3]. Pertuzumab is also licensed for metastatic breast cancer treatment; however, its use for this indication was not considered within this appraisal.

HER2-positive breast cancer is associated with a significantly worse prognosis and higher recurrence rate than other breast cancers [4]. It accounts for around 15% of all breast cancers [4]. Neoadjuvant therapy is sometimes indicated for locally advanced or inflammatory cancer to facilitate or permit surgery in previously inoperable disease. It may be indicated in early breast cancer to facilitate breast conservation surgery or ensure early systemic treatment if surgery may be complicated, such as when reconstruction is planned or to enable gene test results to become available, which may impact on treatment planning. In most cases, neoadjuvant therapy is only advised in women who are at sufficiently high risk of recurrence that they would need adjuvant systemic therapy post operatively [5].

Based on published evidence, the company (Roche Products Limited) estimates that each year there are 5,113 patients with newly diagnosed HER2-positive breast cancer in England. Of these, 27% of patients (based on the company's market research data) would receive neoadjuvant therapy, resulting in 1,380 newly eligible patients per annum for treatment with pertuzumab.

Pertuzumab is given by intravenous infusion. The recommended first dose is 840mg, followed by a dose of 420mg every three weeks until the patient undergoes surgery [3], which will usually be between 3 and 6 doses.

The comparator described within the company's statement of the decision problem is 'neoadjuvant trastuzumab in combination with chemotherapy', whilst the comparator within the final NICE scope was more broadly described as 'standard neoadjuvant therapy without pertuzumab for HER-2 positive breast cancer'. NICE guidance does not currently recommend which treatments to provide within the neoadjuvant setting for these patients. Whilst the company's market research and the ERG's clinical experts suggest that most patients in England would be given the combination regimen of 5-fluorouracil, epirubicin and cyclophosphamide (FEC) followed by a taxane alongside trastuzumab, some patients may receive alternative therapies. Given that no evidence about clinical and cost-effectiveness was provided by the company for patients who would not receive trastuzumab, this assessment was limited to those patients who will receive trastuzumab as neoadjuvant therapy, which the company state is the relevant patient population. It should be noted that whilst the comparator has not been evaluated in terms of cost-effectiveness in the neoadjuvant setting, clinical advisors to the ERG suggest that trastuzumab as adjuvant therapy (which has been assessed and recommended by NICE [6]) has simply been moved to an earlier stage in the patient pathway.

The primary outcome considered was pathological complete response (pCR). Whilst evidence relating to overall survival (OS) and disease-free survival (DFS) was included within the company submission, the key clinical studies were not powered to assess these. Adverse events were reported. The health economic outcome employed within the company's health economic model was the incremental cost per quality-adjusted life year gained, as set out within the NICE Reference Case [1].

# 3. The Independent ERG Review

The company provided a submission to NICE on the clinical and cost-effectiveness of pertuzumab for treating HER2-positive breast cancer at high risk of recurrence (locally advanced, inflammatory or early stage breast cancer, T2/3 or N1), before the patient undergoes surgery, to be used in combination with trastuzumab and chemotherapy [2]. This submission was critically appraised by the ERG. In addition, the ERG identified areas requiring clarification, for which the company provided additional evidence prior to completion of the ERG report [2]. The ERG also modified the company's decision analytic model to produce an ERG base case assessment of cost-effectiveness and to assess the impact of alternative parameter values and assumptions on the model results. This section summarises the evidence presented in the company's submission and the ERG's review of that evidence.

# 3.1 Clinical Evidence provided by the Company

The company submission included a systematic review of the clinical evidence of neoadjuvant pertuzumab. The main supporting evidence was derived from two company-sponsored, multi-country, multi-centre, randomised, open-label, active controlled studies (NeoSphere and TRYPHAENA) [7, 8] assessing the efficacy and safety of neoadjuvant pertuzumab in combination with trastuzumab and chemotherapy for the treatment of HER2-positive early breast cancer.

#### 3.1.1 Clinical study design

The NeoSphere study (a proof of concept study) was a four arm, Phase II trial that randomised 417 treatmentnaïve women, (aged over 18 years) with operable, locally advanced or inflammatory centrally confirmed HER2positive breast cancer (primary tumours >2 cm in diameter) to receive four neoadjuvant cycles of trastuzumab plus docetaxel (Arm A, n=107); pertuzumab plus trastuzumab plus docetaxel (Arm B, n=107); pertuzumab plus trastuzumab (Arm C, n=107) or pertuzumab plus docetaxel (Arm D, n=96) [7]. Pertuzumab was administered at a loading dose of 840mg, followed by a 420mg dose every 3 weeks. Trastuzumab was administered at a loading dose of 8mg/kg, followed by a 6mg/kg dose every 3 weeks. Docetaxel was administered at a dose of 75mg/m<sup>2</sup> (with escalation to 100mg/m<sup>2</sup>, if tolerated) every 3 weeks. Following surgery, all patients received three cycles of adjuvant chemotherapy with the FEC regimen (5-fluorouracil, 600mg/m<sup>2</sup>; epirubicin, 90mg/m<sup>2</sup>; and cyclophosphamide, 600mg/m<sup>2</sup> administered intravenously every 3 weeks) and trastuzumab every 3 weeks to complete 1 year of therapy. Postoperative loco-regional radiotherapy and endocrine treatments for oestrogen receptor positive tumours were given according to local and national guidelines. The primary endpoint was pathological complete response (pCR) in the breast (bpCR), defined in the study as absence of invasive tumour in the breast irrespective of ductal carcinoma in-situ or nodal involvement, ypT0/Tis. Total pathological complete response (tpCR) was also reported, defined in the study as absence of invasive tumour in breast and lymph nodes irrespective of ductal carcinoma in-situ, ypT0/is ypN0. The marketing authorisation for pertuzumab in the neoadjuvant setting is restricted to use in combination with trastuzumab and chemotherapy only; hence, Arm C and Arm D of the NeoSphere study were not considered relevant to the appraisal by the ERG.

The TRYPHAENA study (a cardiac safety study) was a Phase II study that randomised 225 treatment naïve women, (aged over 18 years) with operable, locally advanced or inflammatory centrally confirmed HER2positive breast cancer (primary tumours > 2cm in diameter) to receive one of three neoadjuvant treatments: Arm A (n=73) included pertuzumab and trastuzumab in cycles 1 to 6 plus FEC (5-fluorouracil, 500mg/m<sup>2</sup>; epirubicin, 100mg/m<sup>2</sup> and cyclophosphamide 500mg/m<sup>2</sup>) in cycles 1 to 3 and docetaxel (75mg/m<sup>2</sup> increased to 100mg/m<sup>2</sup> if tolerated) in cycles 4 to 6; Arm B (n=75) included FEC alone in cycles 1 to 3 followed by pertuzumab, trastuzumab and docetaxel (75mg/m<sup>2</sup> increased to 100mg/m<sup>2</sup> if tolerated) in cycles 4 to 6; Arm C (n=77) included pertuzumab, trastuzumab, docetaxel (75mg/m<sup>2</sup> with no dose escalation) and carboplatin in cycles 1 to 6 [8]. Pertuzumab was given at an initial dose of 840mg, with subsequent doses of 420mg. Trastuzumab was given at an initial loading dose of 8mg/kg, followed by 6mg/kg. All regimens were given intravenously every 3 weeks for a total of six neoadjuvant cycles. Following surgery, all patients received trastuzumab every 3 weeks to complete 1 year of therapy. Postoperative loco-regional radiotherapy and endocrine treatments for oestrogen receptor positive tumours were given according to local and national guidelines. The primary endpoint of the study was cardiac safety. The statistical analysis plan did not include any pre-planned hypothesis testing and the submission did not include any statistical comparisons between the treatment arms for any outcome. In addition, because all groups in this study received pertuzumab, comparative efficacy of pertuzumab in combination with trastuzumab and chemotherapy versus trastuzumab and chemotherapy without pertuzumab cannot be estimated using this study.

## 3.1.2 Clinical study results

# Clinical effectiveness

In general, the bpCR rate (trial definition of pCR) in the NeoSphere study was significantly higher in Arm B (combination of pertuzumab, trastuzumab and docetaxel, 45.8%) compared with Arm A (combination of trastuzumab plus docetaxel, 29.0%), with a difference of 16.8% (p=0.0141) [7]. The rates of tpCR (EMA and FDA preferred definition of pCR) was broadly similar to that of bpCR (Arm B, 39.3% versus Arm A, 21.5%; difference of 17.8%, p=0.0063). In the TRYPHAENA study, bpCR and tpCR were consistently high and similar across all treatment groups (approximately 60%) [8].

Although the NeoSphere study was not powered to assess long-term outcomes or subgroups, 5-year progression-free survival (PFS) was 86% for Arm B (95% CI: 77% to 91%) compared with 81% (95% CI: 71% to 87%) for Arm A [7]. The hazard ratio for PFS for Arm B versus Arm A was 0.69 (95% CI: 0.34 to 1.40). The 5-year disease-free survival (DFS) data were 81% and 84% in Arms A and B respectively. The DFS hazard ratio for Arm B versus Arm A was 0.60 (95% CI: 0.28 to 1.27) [7]. In the TRYPHAENA study, DFS data was not yet available at the time of company submission. Data relating to health related quality-of-life (HRQoL) were not collected in either study [8]. Table 1 shows the summary outcomes from the relevant arms of the NeoSphere trial.

# Safety

During the neoadjuvant period of the NeoSphere (<3% across all arms) and TRYPHAENA studies (<8% across all arms), adverse events leading to treatment discontinuation were low [7]. In the neoadjuvant phase of the NeoSphere study, grade ≥3 neutropenia was numerically higher in patients who received docetaxel (Arm A, 57.0%; Arm B, 44.9%; Arm D, 55.3%) than in patients who did not receive docetaxel (Arm C, 1%). The other most common grade ≥3 adverse events were febrile neutropenia (range 7.4% to 8.4% in docetaxel arms and none in the arm without docetaxel) and leucopenia (range 5% to 12% in the docetaxel arms and none in the arm without docetaxel) [7]. There was 1 death possibly related to treatment, in the dual treatment arm (B) in the NEOSPHERE study from fulminant hepatitis. In the TRYPHAENA study, similar incidences of grade ≥3 adverse events were observed (neutropenia, range 46.1% to 47.2%; febrile neutropenia, range 9.3% to 18.1%; leucopenia, range 11.8% to 19.4%) [8]. In the NEOSPHERE study, the number of patients with cardiac dysfunction adverse events was low in all trial arms; this was highest in Arm B (3% to 6% across the treatment periods). Similarly, in the TRYPHAENA study, incidence of symptomatic left ventricular systolic dysfunction (LVSD) and significant declines in left ventricular ejection fraction (LVEF) (≥10% points from baseline to <50%) were low across all arms but highest in Arm B (1.3% to 12.3% across the treatment periods) [8].

# 3.2 Critique of the Clinical Evidence and Interpretation

# 3.2.1 Critique of systematic review

The systematic review process followed by the company was reasonably comprehensive. Despite minor limitations in the company's search strategy, the ERG is confident that all relevant controlled studies of pertuzumab in combination with trastuzumab and chemotherapy for the treatment of HER2-positive early breast cancer were included in the company submission, including data from ongoing or planned studies. However, the ERG is not confident that all relevant non-randomised and non-controlled studies have been identified and

included in the company submission, as details of the systematic review process (e.g. identification, selection, data extraction, quality assessment and analysis and interpretation) were lacking. The specified inclusion and exclusion criteria were mostly appropriate and generally reflect the decision problem set out in the final NICE scope. The validity assessment tool used to appraise the included studies (NeoSphere and TRYPHAENA [7, 8]) was considered appropriate by the ERG.

# 3.2.2 Critique of clinical evidence

Although the efficacy (measured in terms of pCR response) and safety of pertuzumab in combination with trastuzumab and chemotherapy compared with trastuzumab and chemotherapy was positively demonstrated in the key included studies, there are a number of limitations and uncertainties in the evidence base which warrant caution in its interpretation.

## Limitations of the RCTs

The main evidence for clinical efficacy and safety of pertuzumab in the company submission was derived from two, Phase II, randomised, open-label, active controlled studies [7, 8]. There was no evidence from a RCT powered adequately for DFS and OS endpoints. As with many cytotoxic cancer drugs, the nature of the interventions precludes blinding and is almost universally absent from oncology trials; however, blinded outcome assessment can enhance bias reduction. The TRYPHAENA study, which was a cardiac safety study, included pertuzumab in all arms and did not provide evidence of comparative efficacy with treatments without pertuzumab [8]. In addition, because of current practice varying between countries, the generalisability of the results from the RCTs to England is unclear.

### pCR as a surrogate endpoint

The FDA and EMA both granted approval of neoadjuvant pertuzumab on acceptance of pCR as a surrogate endpoint in neoadjuvant treatment for high risk early stage breast cancer based upon work by Cortazar et al.,[9] subject to the need to collect long-term clinical outcomes data [10]. Cortazar et al. performed a patient-level responder analysis and a study-level analysis to investigate the relationship between pCR and both EFS and OS [9]. The authors identified 12 neoadjuvant studies, including 11,955 patients in the responder analysis and 9,440 patients in the study-level analysis. In all patients, this analysis suggested that patients who achieved pCR defined as absence of invasive cancer in the breast and axillary nodes had an improved EFS (Hazard Ratio (HR) 0.48 95% CI: 0.43, 0.54) and OS (HR 0.36 95% CI: 0.31, 0.42) compared to those who did not have a pCR. The greatest association was in patients with HER2-positive, hormone receptor-negative tumours who received trastuzumab (EFS: HR 0.15 95% CI: 0.09, 0.27; OS: HR 0.08 95% CI: 0.03, 0.22) and those in the triplenegative subgroup. However, the analysis was unable to demonstrate a relationship between the effect of treatment on pCR (estimated using an odds ratio) and the effect of treatment on EFS and OS (estimated using a hazard ratio) at the study level. These findings are generally consistent with other similar studies [11, 12]. The ERG accepts that there is evidence at the patient-level that a pCR responder is associated with a lower risk of EFS and OS. However, the evidence that a positive treatment effect on pCR translates into a positive effect on OS is not convincing. Therefore, the predictive value of pCR for estimating the long-term survival benefit in the target patient population is highly uncertain.

# 3.3 Cost-Effectiveness Evidence submitted by the Company

The company identified one existing economic evaluation of pertuzumab for early stage breast cancer in their economic review. This was developed by the company and is similar to the model within the company submission. The company undertook model-based economic evaluation of neoadjuvant pertuzumab plus trastuzumab and docetaxel compared with neoadjuvant trastuzumab and docetaxel over a lifetime horizon from the NHS and PSS perspective. The company's *de novo* model adopts a cohort level state transition approach based on six health states: event-free, locoregional recurrence, remission, metastatic not-progressed, metastatic progressed and death, as shown in Figure 1. Costs and outcomes are evaluated using a monthly cycle length. Patients in the event-free state can transit to locoregional recurrence, the metastatic not-progressed state or death. Patients spend 12 months in locoregional recurrence (which is modelled as a tunnel state without the possibility of transitioning to death), after which they transition to the remission state. Patients in the remission state can transition to the metastatic not-progressed state or death. Patients in the metastatic progressed state can transition to the metastatic progressed state or death. Patients in the metastatic progressed state can transition only to death.

Although the company used the term DFS within the clinical section of the company submission, they used event-free survival (EFS) within the cost-effectiveness section. The ERG believes that the company's intention was that these terms be considered synonymous. Given the limited number of EFS events within the key clinical studies, the company used results from a meta-analysis of 12 neoadjuvant studies investigating the relationship between pCR and EFS by the Collaborative Trials in Neoadjuvant Breast Cancer (CTNeoBC) group [9] to extrapolate the pCR outcomes reported in the NeoSphere trial. The company digitised the EFS Kaplan-Meier survivor functions from the CTNeoBC meta-analysis and used it to reconstruct the data for patients achieving pCR or not achieving pCR, using the algorithm reported by Guyot et al [13]. The company fitted a number of parametric survivor functions (exponential, Weibull, log logistic, log normal, Gompertz and Generalised gamma) to the reconstructed data, and assessed the best fit using visual inspection and Akaike information criterion (AIC)/ Bayesian information criterion (BIC) statistics. In the base case analysis, the company used gamma distributions for patients achieving pCR and for patients not achieving pCR. The EFS for each treatment arm in the model was then estimated by multiplying the survivor functions of pCR and no pCR by the proportions of patients achieving pCR and no pCR in the respective arms in the NeoSphere trial.

Within the model it was assumed that the treatment effect persists for seven years. The company also assumed that seven years after treatment initiation, patients who have not experienced locoregional or metastatic recurrence are assumed to be cured, with only the risk of general population mortality. A utility was assigned to the 'event-free', 'locoregional', and 'metastatic non-progressed' health states based on a study of 361 consecutive breast cancer patients attending an outpatient clinic in Stockholm between April and May 2005 by Lidgren *et al.*[14] The utility value for the metastatic progressed health state was informed by a mixed model analysis by Lloyd *et al.*[15] Dis-utilities for adverse events were not applied. The model includes costs associated with drug acquisition based on the British National Formulary[16] for pertuzumab and trastuzumab and the Commercial Medicines Unit 2014 electronic Market Information Tool[17] for generic drugs. It also

includes costs associated with drug administration, the treatment of adverse events occurring in more than 5% of patients in either arm of the NeoSphere trial at grade 3, 4 or 5 severity and supportive care (all based on Personal Social Services Research Unit (PSSRU) 2014 costs[18] and NHS Reference Costs 2013/14[19]). Costs and health outcomes were discounted at 3.5% per annum.

The company reported a probabilistic ICER within their original submission of £20,104 per QALY gained for pertuzumab alongside trastuzumab and docetaxel compared with trastuzumab and docetaxel. The company amended their base case during the clarification process to £21,869 per QALY gained, by costing trastuzumab using the split in usage of infusional and subcutaneous formulations of trastuzumab based on their market research data. After the clarification process, the ERG highlighted an error around the digitised survivor functions which resulted in a new substantially reduced company probabilistic ICER of £9,047 per QALY gained for pertuzumab alongside trastuzumab and docetaxel compared with trastuzumab and docetaxel. The one-way sensitivity analysis performed by the company suggests that the key driver of the model results is the pCR rates.

# 3.3.1 Critique of the Cost-Effectiveness Evidence and Interpretation

The *de novo* model developed is generally appropriate for the decision problem defined in the final scope, although not all possible comparators have been included. The perspective, outcomes, discount rate, and measurement and valuation of costs and outcomes adhere to the NICE Reference Case. The model was generally well described within the company submission.

The key area of uncertainty concerned the validity of pCR as a surrogate endpoint for EFS and the approach for extrapolation. Although pCR has been used as a surrogate outcome for regulatory approval [10], the ERG has concerns about the use of pCR as a predictor of EFS. The CTNeoBC analyses found a correlation between pCR and EFS at the individual level, but could not validate pCR as a surrogate endpoint for improved EFS at the study-level [9]. The choice of parametric distribution used for extrapolation is shown within sensitivity analysis to impact upon the model results substantially and there is limited longer term data within this patient population to be able to satisfactorily validate this choice. The use of pCR as a surrogate outcome to predict EFS within the health economic model is a poor predictor of the EFS within the NeoSpere trial, irrespective of which parametric distribution is chosen. However, the company did also undertake an analysis using the EFS data directly from the NeoSphere trial which suggested that pertuzumab dominates (i.e. is more effective and less costly than the comparator).

The ERG also identified an error in the digitisation of the survivor function from the CTNeoBC meta-analysis. In the model, the company appeared to have used the data from all breast cancer patients up until around nine years (the length of follow up available for the HER2-positive subgroup, rather than the 18 year follow up available for all breast cancer patients), alongside the numbers at risk from the HER2-positive subgroup. The company and the ERG corrected this error by using only the data from the HER2-positive subgroup.

The justification for the assumption that after seven years from treatment initiation, patients who have not experienced locoregional or metastatic recurrence are assumed to be cured was unclear. Clinical advisors to the ERG suggested that whilst this may be reasonable for the hormone receptor (HR)-negative group, HR-positive patients are likely to continue to experience events and have greater mortality beyond seven years following treatment initiation compared with the general population [20]. Since the clinical advisors to the ERG suggested that cure of all HR-positive patients after 7 years is not clinically plausible, this assumption was amended in the ERG's base case analysis (see Section 3.4).

In addition, the uncertainty around the model parameters for the PSA was inadequately characterised. Within the company submission, the distributions and parameters used in the PSA were neither presented nor justified. Some uncertain model parameters were not characterised by probability distributions, and where included, the characterisation of uncertainty surrounding some model parameters appeared arbitrary. For example, the parameterisation of uncertainty in adverse event cost, administration cost, pharmacy time required for intravenous preparation and supportive care cost was assumed to be a proportion (typically 10-25%) of the mean. In addition, tabled results of the PSA were not presented by the company. Therefore, a clear analysis of uncertainty was not presented.

# 3.4 Additional Work Undertaken by the ERG

The ERG produced a revised base case which was similar to the company's base case following the clarification process. The ERG corrected the error in the digitisation of the survivor functions and modified the clinically inappropriate assumption that the probability of recurrence is zero after seven years. Whilst these changes individually impacted upon the ICER substantially, they acted in different directions and, when incorporated together, did not have a substantial impact on the company's results. The ERG-preferred probabilistic ICER for pertuzumab in combination with trastuzumab plus docetaxel compared with trastuzumab plus docetaxel is estimated to be £23,962 per QALY gained. Similarly, the ERG's deterministic base case ICER is estimated to be £23,467 per QALY gained. Whilst the ERG did not have sufficient time to improve the PSA, the more extensive univariate sensitivity analysis undertaken by the ERG suggested that the key drivers of the model results are: the relative pCR rates associated with the interventions; the parametric distribution employed for extrapolation of EFS; the number of cycles of pertuzumab administered; the costs of second line metastatic treatment; whether the treatment effect is assumed to continue beyond the trial follow-up duration; and health utility values.

# 3.5 Conclusions of the ERG Report

The ERG considered the efficacy (in terms of pCR response [using various definitions]) and safety evidence of pertuzumab in combination with trastuzumab and chemotherapy to be positively demonstrated (compared with trastuzumab and chemotherapy) in the key included studies. However, there are a number of limitations and uncertainties in the evidence base for cost-effectiveness that warrant caution in its interpretation. Treatment effects may be confounded because of the open-label design of the phase II studies. The key uncertainties in the evidence base relate to the use of pCR as a surrogate endpoint for survival outcomes (including magnitude of

benefit), the lack of results from high quality phase III RCTs, and the generalisability of the study results to England and the NHS.

The de novo model developed was generally appropriate for the decision problem defined in the final NICE scope, though it should be noted that the only comparator tested within the economic evaluation was trastuzumab alongside docetaxel. The model structure was considered by the ERG to be reasonable; however, there are uncertainties associated with the use of pCR as a surrogate measure for EFS and it does not appear to be a good predictor of the EFS data from the NeoSphere trial. The company's probabilistic ICER using this surrogate outcome was £20,104 per QALY gained for pertuzumab alongside trastuzumab and docetaxel compared with trastuzumab and docetaxel. The company amended their base case during the clarification process to £21,869 per QALY gained, by costing trastuzumab using the split in usage of infusional and subcutaneous formulations of trastuzumab based on their market research data. In addition to this change, the ERG have corrected an error in the digitisation of the survivor functions and modified the clinically inappropriate assumption that recurrence is zero after 7 years, resulting in a probabilistic ICER of £23,962 per QALY gained for pertuzumab alongside trastuzumab and docetaxel compared with trastuzumab and docetaxel. An alternative analysis was undertaken by the company using the EFS data from the NeoSphere study directly within the analysis, which suggested that pertuzumab, trastuzumab and docetaxel dominates (i.e. is more effective and less costly) compared with trastuzumab and docetaxel alone. The univariate sensitivity analysis undertaken by the ERG suggested that the key drivers of the model results are: the relative pCR rates associated with the interventions; the parametric distribution employed for extrapolation of EFS; the number of cycles of pertuzumab administered; the costs of second line metastatic treatment; whether the treatment effect is assumed to continue beyond the trial follow-up duration; and health utility values.

# 4. Key Methodological Issues

# 4.1 Short term data and the use of surrogate outcomes

NeoSphere was designed to demonstrate efficacy using pCR rates to enable accelerated use. Several studies have attempted to assess the relationship between pCR and event-free and overall survival. pCR was accepted for accelerated approval by both the European and US licensing authorities as a valid and meaningful clinical endpoint for regulatory approval of neoadjuvant breast cancer studies, subject to the need to collect long-term clinical outcomes data [10]. However, there remains uncertainty around whether an effect on pCR translates into effects on survival outcomes. There may be a difference between the evidence requirements for regulatory approval of a health technology and those for making funding decisions in England.

# 4.2 The impact of metastatic treatment costs and the Cancer Drugs Fund upon the model results

A key driver of the health economic model results was the assumption about which treatments for metastatic disease would be provided to patients. Costs of metastatic treatment would be incurred in both the intervention and comparator groups; however, a higher cost of metastatic treatment is favourable for the intervention (pertuzumab). This is because fewer patients would receive metastatic treatment in the intervention group since patients are not progressing so quickly and hence are more likely to die before metastatic progression. In

addition, discounting leads to reduced metastatic costs when incurred later. Thus, costs of subsequent treatment can substantially impact upon the cost-effectiveness of the intervention being assessed.

Due to the Cancer Drugs Fund (CDF), more expensive health technologies have been routinely available to patients than NICE have considered as a cost-effective use of NHS resources. At the time of this appraisal, the use of the CDF was being modified and NICE were in the process of reviewing the cost-effectiveness of all treatments available to patients using the fund. The timing of these reviews meant that it was unknown which metastatic treatments would become standard practice in England during the appraisal. Moreover, if the treatments currently available on the CDF were recommended they would likely be subject to a PAS and hence this would result in a less favourable ICER for pertuzumab. The cost of the metastatic treatments was shown, within sensitivity analysis, to impact substantially upon the model results. To mitigate the uncertainty associated with this, the company proposed a PAS for pertuzumab based upon a pessimistic scenario that neither of the health technologies for metastatic breast cancer which were available on the CDF would be funded following the review by NICE.

# 4.3 Sensitivity analyses

Given the substantial uncertainties associated with the evidence and current metastatic practice, it is important for the appraisal committee to have a clear understanding of the uncertainty which has and has not been incorporated into the model and its impact upon the model results. As described previously, the uncertainty was inadequately characterised within the PSA and insufficient information was provided within the company submission about both the PSA inputs and outputs. In addition, a univariate sensitivity analysis can help the committee to understand the impact of key parameters within the model, and the company did not assess many of the key drivers of the model results such as the number of cycles of pertuzumab and the cost of treating metastatic disease within their submission.

# 5. NICE Guidance

A PAS to discount the price of pertuzumab was proposed by the company following a decision not to recommend neoadjuvant pertuzumab resulting from the first appraisal committee meeting and reported within the ACD. The negative recommendation was because of the substantial uncertainty around the ICER, mainly associated with the use of pCR as a surrogate marker for long term outcomes and whether CDF-funded treatments for metastatic disease which are or will be under review should be included within the analysis. The company therefore proposed a PAS to mitigate the risk associated with recommending pertuzumab, the level of discount being designated as commercial in confidence. When the company used the ERG's assumptions and made the conservative assumption that treatment of metastatic disease did not include CDF-funded treatments, with the incorporation of the PAS, the ICER fell within the range normally considered to be a cost-effective use of NHS resources. The ERG was not asked to undertake any additional analysis around the PAS. In November 2016, NICE published its Final Appraisal Determination (FAD), which states that pertuzumab, in combination with trastuzumab and chemotherapy, is recommended as an option for the neoadjuvant treatment of HER2-positive breast cancer at high risk of recurrence, only if the company provides pertuzumab with the discount agreed in the PAS. Patients should normally have no more than four cycles.

#### 6. Conclusions

The evidence suggests that pertuzumab is an effective and safe option for the neoadjuvant treatment of patients with early stage HER2-positive breast cancer at high risk of recurrence, to be used together with trastuzumab and chemotherapy. However, uncertainty remains around the extent to which short term improvements in pCR translate to long term gains in survival. The estimated base case ICER reported by both the company and the ERG fell below £30,000 per QALY gained compared with trastuzumab and docetaxel; although there was substantial uncertainty around this estimate. To mitigate this uncertainty, a PAS was proposed by the company, which allowed NICE to recommend pertuzumab for this indication as an expected cost-effective use of NHS resources.

# Acknowledgements

## **Author Contributions**

HS drafted the manuscript and takes responsibility as guarantor of the content. AP, PT, JS, EK, MC, RC and LW revised the manuscript for important intellectual content. All authors have given their approval for the final version to be published. This summary has not been externally reviewed by PharmacoEconomics.

# **Compliance with Ethical Standards**

**Funding** 

This project was funded by the National Institute for Health Research (NIHR) Health Technology Assessment Programme (project number 15/69/11 STA). See the HTA programme website for further project information (<a href="http://www.hta.ac.uk">http://www.hta.ac.uk</a>). This summary of the ERG report was compiled after NICE issued guidance. The views and opinions expressed therein are those of the authors and do not necessarily reflect those of NICE or the Department of Health.

Conflicts of interest

HS, AP, PT, JS, EK, MC, RC and LW have no potential conflicts of interest.

## References

- 1. **NICE technology appraisal guidance** [https://www.nice.org.uk/About/What-we-do/Our-Programmes/NICE-guidance/NICE-technology-appraisal-guidance]
- 2. Breast cancer (HER2 positive) pertuzumab (neoadjuvant) [ID767] [https://www.nice.org.uk/guidance/indevelopment/gid-ta10011]
- Annex I Summary of product characteristics. Perjeta EPAR
   [http://www.ema.europa.eu/ema/index.jsp?curl=pages/medicines/human/medicines/0025
   47/human med 001628.jsp&mid=WC0b01ac058001d124]
- 4. Single Technology Appraisal Pertuzumab for the neoadjuvant treatment of HER2-positive breast cancer. Final scope [https://www.nice.org.uk/guidance/ta424]
- 5. Kaufmann M. vMG, Bear H.D., Buzdar A., McGale P., Bonnefoi H., Colleoni M., Denkert C., et al.: Recommendations from an international expert panel on the use of neoadjuvant (primary) systemic treatment of operable breast cancer: new perspectives 2006. *Ann Oncol* 2007, 18:1927-1934.
- 6. Advanced breast cancer: diagnosis and treatment [CG81] [https://www.nice.org.uk/guidance/cg81]
- 7. Gianni L, Pienkowski T, Im YH, Roman L, Tseng LM, Liu MC, Lluch A, Staroslawska E, de la Haba-Rodriguez J, Im SA, et al: Efficacy and safety of neoadjuvant pertuzumab and trastuzumab in women with locally advanced, inflammatory, or early HER2-positive breast cancer (NeoSphere): a randomised multicentre, open-label, phase 2 trial. *Lancet Oncol* 2012, 13:25-32.
- 8. Schneeweiss A, Chia S, Hickish T, Harvey V, Eniu A, Hegg R, Tausch C, Seo JH, Tsai YF, Ratnayake J, et al: Pertuzumab plus trastuzumab in combination with standard neoadjuvant anthracycline-containing and anthracycline-free chemotherapy regimens in patients with HER2-positive early breast cancer: a randomized phase II cardiac safety study (TRYPHAENA). Annals of Oncology 2013, 24:2278-2284.
- 9. Cortazar P, Zhang L, Untch M, Mehta K, Costantino JP, Wolmark N, Bonnefoi H, Cameron D, Gianni L, Valagussa P, et al: **Pathological complete response and long-term clinical benefit in breast cancer: the CTNeoBC pooled analysis.** *Lancet* 2014, **384**:164-172.
- 10. European Medicines Agency (EMA): **Assessment Report For Perjeta (pertuzumab) Procedure No: EMEA/H/C/002547/II/0010. 2015.** 2015.
- 11. Berruti A, Amoroso V, Gallo F, Bertaglia V, Simoncini E, Pedersini R, Ferrari L, Bottini A, Bruzzi P, Sormani MP: Pathologic complete response as a potential surrogate for the clinical outcome in patients with breast cancer after neoadjuvant therapy: a meta-regression of 29 randomized prospective studies. *J Clin Oncol* 2014, 32:3883-3891.
- 12. Broglio KR, Quintana M, Foster M, Olinger M, McGlothlin A, Berry SM, Boileau JF, Brezden-Masley C, Chia S, Dent S, et al: Association of Pathologic Complete Response to Neoadjuvant Therapy in HER2-Positive Breast Cancer With Long-Term Outcomes: A Meta-Analysis. *JAMA Oncol* 2016.
- 13. Guyot P, Ades A, Ouwens MJ, Welton NJ: Enhanced secondary analysis of survival data: reconstructing the data from published Kaplan-Meier survival curves. *BMC Medical Research Methodology* 2012, **12**:1-13.
- 14. Lidgren M, Wilking N, Jonsson B, Rehnberg C: **Health related quality of life in different states of breast cancer.** *Qual Life Res* 2007, **16**:1073-1081.
- 15. Lloyd A, Nafees B, Narewska J, Dewilde S, Watkins J: **Health state utilities for metastatic breast cancer.** *Br J Cancer* 2006, **95**:683-690.
- 16. British National Formulary (BNF) [www.bnf.org/]
- 17. Commercial Medicines Unit (CMU): **Drugs and pharmaceutical electronic market information (eMit).** 27/11/2015 edition; 2015.
- 18. Unit PSSR: **Unit Costs of Health and Social Care.** Kent: PSSRU; 2014.

- 19. **NHS reference costs 2013 to 2014** [https://www.gov.uk/government/publications/nhs-reference-costs-2013-to-2014]
- 20. Perez EA, Romond EH, Suman VJ, Jeong J-H, Sledge G, Geyer CE, Martino S, Rastogi P, Gralow J, Swain SM, et al: **Trastuzumab Plus Adjuvant Chemotherapy for Human Epidermal Growth Factor Receptor 2–Positive Breast Cancer: Planned Joint Analysis of Overall Survival From NSABP B-31 and NCCTG N9831.** *Journal of Clinical Oncology* 2014.